Author’s response to reviews

Title: A cerebral Arteriovenous Malformation mistakenly diagnosed as Dry Eye and Glaucoma: a case report

Authors:
Shuhao Shen (jnussh@hotmail.com)
Xiaoyong Liu (jndxlxy@163.com)
Jian Chen (drchenj@163.com)
Chengyou Yang (835013859@qq.com)
Changzheng Shi (tsczcn@jnu.edu.cn)
Qing Zhou (kerryzh@163.com)

Version: 2 Date: 02 Jul 2019

Author’s response to reviews:

Dear reviewers and editors:

First of all, I would like to thank you for your recognition of our work and your acceptance of the case report. Without your valuable comments, this article could not have been successfully completed and accepted by BMC Ophthalmology. According to the opinions of reviewers, I will explain the content modification I have made:

1. According to the opinions of reviewers, I combined the "Discussion" and "Conclusion" sections as "Discussion and conclusion", and represent authors’ names using their initials in section "Author contributions".

2. In the last version, I added tests to support the diagnosis of the disease, such as fundus images, so that readers can know more about the diagnosis of the disease.

3. The article has been revised and polished by AJE and is believed to be more in line with readers’ reading habits.

Thanks again for your valuable comments.